Author's response to reviews

Title: High tidal volume mechanical ventilation-induced lung injury in rats is greater after acid instillation than after sepsis-induced acute lung injury, but does not increase systemic inflammation: an experimental study.

Authors:

Jan Willem Kuiper (jw.kuiper@vumc.nl)
Frans B Plotz (fb.plotz@planet.nl)
AB Johan Groeneveld (johan.groeneveld@vumc.nl)
Jack J Haitsma (jack.haitsma@utoronto.ca)
Serge J Jothy (jothy@smh.toronto.on.ca)
Rosanna Vaschetto (vaschet@med.unipmn.it)
Haibo Zhang (haibo.zhang@utoronto.ca)
Arthur S Slutsky (arthur.slutsky@utoronto.ca)

Version: 6 Date: 19 December 2011

Author's response to reviews: see over
Dear Sir/Madam,

We herewith resubmit our manuscript with the formatting changes as required. Please note that not all names are exactly in the preferred style. The first author first name is Jan Willem. Also, concerning dr Groeneveld, A.B.(J) are the initials, but the first name is Johan, therefore A.B. Johan Groeneveld.

With kind regards and happy holidays,

Jan Willem Kuiper